Author’s response to reviews

Title: Relationship between risk information on total colonoscopy and patient preferences for colorectal cancer screening options: Analysis using the Analytic Hierarchy Process

Authors:

Yuichi Katsumura (ykatsumura-tyky@umin.ac.jp)
Hideo Yasunaga (yasunagah-tyyo@h.u-tokyo.ac.jp)
Tomoaki Imamura (imamura-t@umin.ac.jp)
Kazuhiko Ohe (kohe@hcc.h.u-tokyo.ac.jp)
Hiroshi Oyama (hoyama-nsu@umin.net)

Version: 5 Date: 16 May 2008

Author’s response to reviews:

Dear Editor,

We have changed the format of our manuscript according to your comments. Thank you.

Sincerely,

Hideo Yasunaga